Author's response to reviews

**Title:** Impact of a multifaceted intervention to improve the clinical management of osteoporosis. The ESOSVAL-F study.

**Authors:**

Jose Sanfélix-Genovés (sanfelix_jos@gva.es)  
Salvador Peiró (peiro_bor@gva.es)  
Gabriel Sanfélix-Gimeno (sanfelix_gab@gva.es)  
Isabel Hurtado (hurtado_isa@gva.es)  
Manuel Pascual de la Torre (pascual_man@gva.es)  
Jose Luis Trillo-Mata (trillo_ilu@gva.es)  
Vicente Giner Ruiz (ginerpatres@wanadoo.es)

**Version:** 2  **Date:** 19 October 2010

**Author's response to reviews:** see over
Formatting changes to be made:

Major revisions (we require the author to make these changes)

Table: Please ensure that this is cited in the manuscript text as 'Table 1'.

We have modified in page 10.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Summary: Please change this to 'Abstract'.

We have modified

Table: Please include the title above the table, not in the table.

We have modified

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Page 3. Abstract

We have added: The study’s primary endpoint is the appropriateness of treatment according by the Spanish National Health System guide (2010) and the National Osteoporosis Foundation (NOF, 2008) and International Osteoporosis Foundation guidance (IOF, 2008).

Page 10. Main endpoint:


References (page 19)

We have changed reference number 14 and added reference number 16.